Author’s response to reviews

Title: Implementing a toolkit for the prevention, management and control of carbapenemase-producing Enterobacteriaceae in English acute hospitals trusts: a qualitative implementation evaluation

Authors:

Annegret Schneider (a.schneider@ucl.ac.uk)

Caroline Coope (caroline.coope@bristol.ac.uk)

Susan Michie (s.michie@ucl.ac.uk)

Richard Puleston (Richard.Puleston@phe.gov.uk)

Susan Hopkins (Susan.Hopkins@phe.gov.uk)

Isabel Oliver (isabel.oliver@phe.gov.uk)

Version: 2 Date: 28 Aug 2019

Author’s response to reviews:

Dear Jashan P. Abraham,

Thank you for the additional comments. We made all required revisions as following:

1. Authors’ contributions section – Authors’ contributions were listed in authorship order.

2. Ethics Approval and Consent to Participate section – We added that informed consent was obtained in writing.

3. Funding section – We added a sentence highlighting that the funding body had no role in the design of the study, data collection, analysis, and interpretation of data, and in writing the manuscript.

4. Corresponding author details - This has been amended in the editorial submission to match the title page.

5. Response to Reviewers - The previous response letter has been removed.

6. Figures/tables citations – We double-checked that all figures and tables are cited within the text.

7. Formatting – The figure was moved from the body of the text to follow the reference section.
8. Figure Legend – The figure caption provided in the body of text was moved to the Figure Legend after the References section.

9. Clean Manuscript - All resubmitted files are clean versions now.

Thank you for your continued interest and consideration of our work. We look forward to hearing from you.

Yours Sincerely,

Annegret Schneider, Caroline Coope, Susan Michie, Richard Puleston, Susan Hopkins, Isabel Oliver